Author's response to reviews

Title: A pragmatic study exploring the prevention of delirium among hospitalized older hip fracture patients: applying evidence to routine clinical practice using clinical decision support.

Authors:

Jayna M Holroyd-Leduc (jayna.holroyd-leduc@albertahealthservices.ca)
Greg A Abelseth (gabel01@telus.net)
Farah Khandwala (farah.khandwala@albertahealthservices.ca)
James L Silvius (james.silvius@albertahealthservices.ca)
David B Hogan (dhogan@ucalgary.ca)
Heidi N Schmaltz (heidi.schmaltz@albertahealthservices.ca)
Cyril B Frank (jcrawfor@ucalgary.ca)
Sharon E Straus (sharon.straus@utoronto.ca)

Version: 4 Date: 19 October 2010

Author's response to reviews:

I have carefully reviewed the manuscript and have accepted all the changes except for one change within the abstract. I have rewritten the following sentence so has to reflect the correct meaning: "A Durbin Watson test was conducted to test for serial correlation and, because no correlation was found, Chi-square statistics, Wilcoxon test and logistic regression analyses were conducted as appropriate".

Additionally, the reference to the CAM tool (Methods section; 3rd paragraph) has been corrected to reference 22 (instead of 21).

Sincerely, Jayna Holroyd-Leduc